NPS ARCHIVE 1997。ひう BOSTON, P.

# NAVAL POSTGRADUATE SCHOOL Monterey, California



# THESIS

### A CASE STUDY OF THE MATERIALS MANAGEMENT DEPARTMENT AT THE NAVAL MEDICAL CENTER SAN DIEGO BENCHMARKING EFFORT

by

Pia S. Boston

March, 1997

Thesis Advisor: Associate Advisor: Linda E. Wargo Erik Jansen

Thesis B72646

Approved for public release; distribution is unlimited.

DUDLEY KNOX LIBRARY NAVAL POSTGRADUATE SCHOOL MONTEREY CA 93943-5101

-

•

## **REPORT DOCUMENTATION PAGE**

Form Approved OMB No. 0704-0188

Public reporting burden for this collection of information is estimated to average 1 hour per response, including the time for reviewing instruction, searching existing data sources, gathering and maintaining the data needed, and completing and reviewing the collection of information. Send comments regarding this burden estimate or any other aspect of this collection of information, including suggestions for reducing this burden, to Washington Headquarters Services, Directorate for Information Operations and Reports, 1215 Jefferson Davis Highway, Suite 1204, Arlington, VA 22202-4302, and to the Office of Management and Budget, Paperwork Reduction Project (0704-0188) Washington DC 20503.

				-										
1.	AGENCY USE ONLY (Leave b	ENCY USE ONLY (Leave blank) 2. REPORT DATE 3. REPO March 1997 Mast							YPE AND DATES COVERED Thesis					
4.	TITLE AND SUBTITLE A C MANAGEMENT DEPAR CENTER SAN DIEGO BE	NDINC	6 NUMBERS											
6.	AUTHOR(S) Pia S. Boston													
<ol> <li>PERFORMING ORGANIZATION NAME(S) AND ADDRESS(ES) Naval Postgraduate School Monterey CA 93943-5000</li> </ol>									AING ZATION NUMBER					
9.	SPONSORING/MONITORING	G AGENO	CY NA	AME(S) AND AD	DRESS(	ES)	10. SP AC	ONSOF GENCY	LING/MONITORING REPORT NUMBER					
11.	11. SUPPLEMENTARY NOTES The views expressed in this thesis are those of the author and do not reflect the official policy or position of the Department of Defense or the U.S. Government.													
12a. DISTRIBUTION/AVAILABILITY STATEMENT       12b. DISTRIBUTION/AVAILABILITY STATEMENT         Approved for public release; distribution is unlimited.       12b. DISTRIBUTION/AVAILABILITY STATEMENT									JTION CODE					
13. ABSTRACT (maximum 200 words) This thesis sought to provide lessons learned, recommendations and provoke thought among medical logisticians on the use of benchmarking. The researcher used a single case research strategy to assess how successful the Materials Management Department at the Naval Medical Center San Diego has been in implementing benchmarking as suggested by strategic objective 2.5.43 of the 1994 draft of the Navy Medical Logistics Strategic Plan. Information on the implementation of benchmarking in the Materials Management Department reviews and direct observation. The research included reading and reviewing the current literature on benchmarking to compare private sector thinking with current practices in the Materials Management Department. The benchmarking case used the Ten-Step Department of the Navy Benchmarking Model and interview questions. The analysis and conclusions are based upon the initial research questions and the framework of the critical success factors for a benchmarking study. The														
14. SUBJECT TERMS: Benchmarking, Materials Management, Case Study								15.	NUMBER OF PAGES 80					
								16.	PRICE CODE					
17.	SECURITY CLASSIFI- CATION OF REPORT Unclassified	. SECU CATI Uncl	RITY ON OI assifi	CLASSIFI- F THIS PAGE ed	19. SE CA Ut	CURITY CL ATION OF A nelassified	CLASSIFI- 20. LIMITATION ABSTRACT ABSTRACT ed UL							

NSN 7540-01-280-5500

Standard Form 298 (Rev. 2-89) Prescribed by ANSI Std 239-18 298-102

ii

Approved for public release; distribution is unlimited.

## A CASE STUDY OF THE MATERIALS MANAGEMENT DEPARTMENT AT THE NAVAL MEDICAL CENTER SAN DIEGO BENCHMARKING EFFORT

Pia S. Boston

Lieutenant, United States Navy B.S., Florida A&M University, 1989

Submitted in partial fulfillment of the requirements for the degree of

### **MASTER OF SCIENCE IN MANAGEMENT**

from the

#### NAVAL POSTGRADUATE SCHOOL

March 1997



BUBLEY KNOX LIBRARY NAVAL POUTGRADUATE SCHOOL MONTEREY CA 93943-5101

#### ABSTRACT

This thesis sought to provide lessons learned, recommendations and provoke thought among medical logisticians on the use of benchmarking. The researcher used a single case research strategy to assess how successful the Materials Management Department at the Naval Medical Center San Diego has been in implementing benchmarking as suggested by strategic objective 2.5.43 of the 1994 draft of the Navy Medical Logistics Strategic Plan. Information on the implementation of benchmarking in the Materials Management Department was based upon a questionnaire, document reviews and direct observation. The research included reading and reviewing the current literature on benchmarking to compare private sector thinking with current practices in the Materials Management Department. The benchmarking case used the Ten-Step Department of the Navy Benchmarking Model. The analysis and conclusions are based upon the initial research questions propositions and the framework of the critical success factors for a benchmarking study. The results of the case suggest a cost-benefit analysis was done to purchase sterilization equipment.



#### TABLE OF CONTENTS

I.	INTRO	DUCTI	on .	•	• • •	• •	•	• •	•	•	•	•	•	•	•		•		1
	Α.	BACK	GROUNI	D.		• •	•	•••		•	•	•	•	•	•	•	٠	٠	1
	в.	OBJE	CTIVE	OF	THES	is .	•		•	•	•	•	٠	•	•	•	•	•	1
	С.	RESE	ARCH (	QUES	STIONS	5.	•	• •	•	•	•	•	•	•	•	٠	•	•	2
		1.	Prima	ary	Quest	tion		• •	•		•	•	•	•	•	•	•	•	2
		2.	Secor	ndaı	ry Que	estic	ons	•	•	•	•	•	•	•	•	•	•	•	2
	D.	SCOP	E AND	LIN	1ITAT I	IONS	•		•	•	•	•		•	•	•	•		2
	Ε.	ORGA	NIZAT	ION	OF TH	HE TH	HES	IS	•	•	•	•	•	•	•		•	•	3
II.	RESE	ARCH	METHOI	DS .				•••		•	•		٠	•	•	•	•	•	5
	Α.	OBJE	CTIVE	OF	THE P	RESEA	ARC	н.	•	•	•	•	•		•	•	•	•	5
	Β.	RESE.	ARCH N	METH	HOD .		•	•••	•	•	•	•		•	•	•	•	•	5
		1.	Data	Gat	cherir	ng .	•		•	•		•	•	•	•	•	•		6
			a.	Lit	teratı	ıre H	Rev	iew	•	•	•	•	٠	۰	•	•	•		6
			b.	Sit	te Vis	sit	•	•••	•	•				•	•	•	•	•	6
			с.	Int	cervie	ews	•		•	٠	•	٠	•	•	•	•			7
			d.	Doc	cument	: Rev	vie	WS	•	٠	•	•	•	•	•	•	•	•	9
			е.	Obs	servat	cions	3	•••	•	•		•	•	٠	•	•	٠	•	9
		2.	Analy	vtid	stra	atea	,												10

III.	LITE	ERATU	RE REVIEW
	Α.	BENC	HMARKING DEFINED
	в.	REAS	ONS FOR BENCHMARKING
	с.	THE I	BENCHMARKING PROCESSES
		1.	Four-Step Benchmarking Model 15
		2.	Xerox Corporation Nine-Step Quality Process
		3.	Ten-Step Department of the Navy (DON)
			Benchmarking Model
	D.	CRIT	ICAL SUCCESS FACTORS FOR BENCHMARKING 24
		1.	Understand Your Own Process First 24
		2.	Make Benchmarking a Part of Strategic Planning
		3.	Leadership Support and Commitment 25
		4.	Education and Training
		5.	Communication is Paramount
		6.	Provide Adequate Resources
			a. Time Commitment
			b. Proper Team Composition
			c. Cost
	Ε.	SUMM	ARY

IV.	THE	CASE	STUDY	
	Α.	BACH	GROUNI	
	в.	THE	STUDY	
		1.	Plan	ning Phase
			a.	Identify What to Benchmark 32
			b.	Identify Benchmarking Partners from Best-
				in-Class
			c.	Determine Data Collection Methods &
				Collect Data
		2.	Do Pł	nase
			a.	Determine Performance Gaps 36
			b.	Take a System's View
		3.	Study	y Phase
			a.	Communicate Benchmarking Findings and
				Gain Acceptance and Support 37
			b.	Functional Goals
		4.	Act I	Phase
			a.	Develop Action Plan
			b.	Implement Best Practice Procedures and
				Monitor Progress
			с.	Recalibrate Benchmarks
	С.	INTE	CRVIEW	RESPONSES
		1.	Bench	nmarking Defined
		2.	Leade	ership Support

<ul> <li>3. Other Organizational Expertise</li></ul>															
<ul> <li>4. TQL Office</li></ul>	3.	Othe	r Orga	nizatio	onal E	xpe	rti	se		•	•	•	•		40
5. Education and Training       4         6. Lessons Learned       4         7. Follow-up       4         8. Rewards and Incentives       4         9. Public Relations       4         9. Public Relations       4         ANALYSIS AND CONCLUSION       4         A. MATERIALS MANAGEMENT DEPARTMENT EXPERIENCE VERSURATION       4         1. Definition of Benchmarking       4         2. Reasons for Benchmarking       4         3. The Ten-Step Department of the Nav Benchmarking Model       4         a. Plan Phase       4         b. Do Phase       4         c. Study Phase       4         4. Critical Success Factors       4         b. Make Benchmarking an Integral Part of Strategic Planning       4	4.	TQL (	Office	• • •	• • •	•						•	•	•	40
6.       Lessons Learned       4         7.       Follow-up       4         8.       Rewards and Incentives       4         9.       Public Relations       4         9.       Public Relations       4         ANALYSIS AND CONCLUSION       4         A.       MATERIALS MANAGEMENT DEPARTMENT EXPERIENCE VERSUL         INDUSTRY THOUGHT       4         1.       Definition of Benchmarking       4         2.       Reasons for Benchmarking       4         3.       The       Ten-Step       Department       6         4.       Plan Phase       4       4       4         4.       Critical Success Factors       4       4         4.       Critical Success Factors       4       4         5.       Make Benchmarking an Integral Part of Strategic Planning       4       4	5.	Educa	ation	and Tra	aining	•		•	•	•	•	•	•	•	40
7. Follow-up       4         8. Rewards and Incentives       4         9. Public Relations       4         9. Public Relations       4         9. Public Relations       4         ANALYSIS AND CONCLUSION       4         A. MATERIALS MANAGEMENT DEPARTMENT EXPERIENCE VERSUL         INDUSTRY THOUGHT       4         1. Definition of Benchmarking       4         2. Reasons for Benchmarking       4         3. The Ten-Step Department of the Nav         Benchmarking Model       4         a. Plan Phase       4         b. Do Phase       4         d. Act Phase       4         4. Critical Success Factors       4         b. Make Benchmarking an Integral Part of Strategic Planning       4	6.	Less	ons Le	arned .		•			•	•	•	•	•		41
<ul> <li>8. Rewards and Incentives</li></ul>	7.	Follo	ow-up			•		•	•	•	•	•	•	•	42
9. Public Relations       4         ANALYSIS AND CONCLUSION       4         A. MATERIALS MANAGEMENT DEPARTMENT EXPERIENCE VERSUL         INDUSTRY THOUGHT       4         1. Definition of Benchmarking       4         2. Reasons for Benchmarking       4         3. The Ten-Step Department of the Nav         Benchmarking Model       4         a. Plan Phase       4         b. Do Phase       4         c. Study Phase       4         4. Critical Success Factors       4         b. Make Benchmarking an Integral Part of Strategic Planning       4	8.	Rewa	rds an	d Incer	ntives		• •		•	•	•		•		42
ANALYSIS AND CONCLUSION	9.	Publ:	ic Rel	ations		•		•	•	•	•	•	•	•	42
<ul> <li>ANALYSIS AND CONCLUSION</li></ul>															
<ul> <li>A. MATERIALS MANAGEMENT DEPARTMENT EXPERIENCE VERSUL INDUSTRY THOUGHT</li></ul>	ANALYSIS	AND CON	NCLUSI	ON							•	•	•	•	43
INDUSTRY THOUGHT	A. MA	FERIALS	MANAG	EMENT	DEPAR	FMEN	1T	EXE	PER	IE	NC	E	VE	RS	SUS
<ol> <li>Definition of Benchmarking</li></ol>	INI	DUSTRY 7	[HOUGH	т		•			•	•	•		•	•	43
<ol> <li>Reasons for Benchmarking</li></ol>	1.	Defin	nition	of Ber	nchmar	kin	g.				•			•	43
3. The Ten-Step Department of the Nav Benchmarking Model	2.	Reaso	ons fo	r Bencł	nmarki	ng	• •				•	•	•	•	44
Benchmarking Model	3.	The	Ten-	Step	Depar	tme	nt	(	of		th	le		Na	avy
<ul> <li>a. Plan Phase</li></ul>		Bencl	nmarki	ng Mode	el	•		•			•	•	•	•	45
<ul> <li>b. Do Phase</li></ul>		a.	Plan	Phase		•		•	•	•	•	•	•	•	45
<ul> <li>c. Study Phase</li></ul>		b.	Do Ph	ase .		•					•	•	•	•	46
<ul> <li>d. Act Phase</li></ul>		с.	Study	Phase				•	•	•				•	47
<ul> <li>4. Critical Success Factors</li></ul>		d.	Act P	hase .		•		-		•	•	•	•		47
a. Understand Your Own Process first 4 b. Make Benchmarking an Integral Part of Strategic Planning 4	4.	Crit	ical S	uccess	Facto	rs			•	•	•	•	•	•	48
b. Make Benchmarking an Integral Part of Strategic Planning 4		a.	Under	stand Y	Your O	wn	Prc	ce	SS	fi	rs	st			48
Strategic Planning 4		b.	Make	Benchm	arkin	g a	n	Int	ceg	ra	1	Ρä	art	t	of
			Strat	egic Pl	lannin	g		•					•		48
c. Leadership Support and Commitment 4		с.	Leade	rship (	Suppor	t a:	nd	Cor	nmi	Ltm	ien	ıt			49

V

x

			d.	Timely	Edi	ucati	on	and		[ra	ini	ng		is
				Import	ant .	• •	• •	• •					•	49
			e.	Commun	icatio	on is	Par	amou	nt	٠	• •			50
			f.	Provid	e Adeo	quate	Res	ourc	es	•		•	•	50
	в.	SUCCE	ESSFUI	L IMPLE	MENTA	FION	OF B	ENCH	MARI	KIN	G.		٠	50
	С.	RECON	MENDA	ATIONS	FOR F	JTURE	RES	EARC	н.			•		52
		1.	Feasi	lbility	of	Bend	chmaı	cking	g i	n	Ma	teı	cia	als
			Manaq	gement	Depar	tment	s in	Nav	y Me	edi	cin	е		52
		2.	Bench	nmarkin	g Com	paris	on S	tudy	•	•	• •	•	•	53
		3.	Asses	ss the	Resu	lts	of	the	In	teg	rat	io	n	of
			Bench	nmarkin	g into	o a S	trat	egic	Pla	an	•••	•		53
	D.	SUMMA	ARY .	• • •		• •	• •	• •		•			•	53
APPE	NDIX.	INTH	ERVIEV	V QUEST	IONS	•••	• •	•••	• •	٠	•••	•		55
LIST	OF RI	EFEREN	NCES	• • •	•••				• •	•	•••	•	•	59
BIBL	IOGRAI	PHY .		• • •	• • •	• •	• •	•••	•••			•	٠	63
INIT	IAL D	ISTRI	SUTION	I LIST					• •			•	•	67

To my parents and siblings, thank you for your continued support and confidence as I pursued a Masters of Science in Materials Logistics. My next educational journey is earning a Ph.D. in the next millennium.

To my dearest friend, Anthony R. Woodley, thank you for your love and support. I know it was tough at times for you to stay away while I worked on my thesis but you managed it with faith and courage.

Finally, a warm thank you is due to Daisy Ware (Grandmother), Sara Bellamy-Walker (Cousin), Bridget Taylor, Alex Michael, and Antionette(Walker) Bell without the knowledge of each of you my life would have been a little tougher. Thank you and Thank God.

xii

#### I. INTRODUCTION

This chapter explains the objective of the thesis, presents the research questions, describes the scope, limitations, and assumptions of the research effort, and delineates the organization of the study.

#### A. BACKGROUND

In 1994 the leaders in the medical logistics community drafted the Navy Medical Logistics Strategic Plan. The initial plan consisted of five goals, several strategies, and an ambitious number of objectives. Strategic objective 2.5.43 suggested benchmarking with other organizations to learn about industry's best practices.

Medical logistics leaders have had a history of establishing progressive goals and strategies. Yet, simply establishing goals and strategies is not enough; implementation, monitoring, and ongoing review need to occur. This thesis will describe how each of these elements transpired in one activity, the Materials Management Department (MMD) of Naval Medical Center (NMC) San Diego, which has completed a benchmarking effort.

Since the introduction of that draft, the final version of the Navy Medical Logistics Strategic Plan has been published. Although, the logistics chiefs did not include strategic objective 2.5.43 in the final plan, benchmarking is still a useful tool to facilitate organizational change. When an organization decides to benchmark, they are seeking a competitive edge, world-class status, and seeking to become a leader in customer satisfaction.

#### B. OBJECTIVE OF THESIS

This thesis provides lessons learned and recommendations, and it provokes thought among medical

logisticians on the use of benchmarking. The author used a single-case method to assess how successful the Materials Management Department at the Naval Medical Center San Diego was at implementing benchmarking as suggested by strategic objective 2.5.43 of the 1994 draft of the Navy Medical Logistics Strategic Plan. Research concentrated on assessing individuals' knowledge and perception of the benchmarking effort, their assessment of its value and identifying potential barriers and lessons learned.

#### C. RESEARCH QUESTIONS

The following research questions are addressed in this thesis:

#### 1. Primary Question

How successful has the Materials Management Department at Naval Medical Center San Diego been in implementing benchmarking as required by strategic objective 2.5.43 of the 1994 draft of the Navy Medical Logistics Strategic Plan?

#### 2. Secondary Questions

What is benchmarking and what are the critical elements for benchmarking success?

How is the Materials Management Department implementing benchmarking?

How does the Materials Management Department experience compare to private sector thought regarding benchmarking?

What are the "lessons learned" from the Materials Management Department experience that may have broader medical logistics community applicability?

#### D. SCOPE AND LIMITATIONS

The scope of this thesis is a case study that consists of three parts: 1) a description of a case, 2) an analytic generalization of the data, and 3) lessons learned from the research. The research was conducted for more than six

months and included reading and the reviewing current literature on benchmarking. Information on the implementation of benchmarking in the Materials Management Department is based upon one questionnaire, document reviews and direct observation of the results of the study. This study is limited to one division in one organization. One should be cautious in generalizing the results to other divisions or organizations within Navy Medicine because many of the original study participants were not questioned. Almost all had either changed duty stations or had retired. This resulted in interviews with individuals indirectly related to the benchmarking effort. In addition, little written historical documentation was available about the benchmarking study. These factors directly affected the level of detail in the case and the depth of the analysis.

#### E. ORGANIZATION OF THE THESIS

This case study assumes that the reader has little knowledge of benchmarking. If the reader is already familiar with benchmarking it may be useful to read this section, then proceed to areas of interest.

Chapter I explains the objective of this thesis, presents the research questions, describes the scope and limitations, and delineates the organization of the study.

Chapter II introduces the research strategy and methods.

Chapter III discusses private sector thought regarding benchmarking.

Chapter IV is a description of the study in the framework of the critical success factors and the Ten-Step Department of the Navy Benchmarking Model.

Chapter V analyzes the findings from the implementation of benchmarking in the Materials Management Department. The

analysis focuses on comparing the case study with private sector thought. It also identifies areas of future research.

#### A. OBJECTIVE OF THE RESEARCH

This research will focus on the implementation of benchmarking in one organization, the Materials Management Department at the Naval Medical Center San Diego. Strategic objective 2.5.43 of the 1994 draft Navy Medical Logistics Strategic Plan suggested benchmarking with other organizations. Although logistics chiefs did not include this objective in the final strategic plan, benchmarking is still a useful tool to facilitate change in an organization and promote continuous process improvement. This thesis assesses one department's success in carrying out benchmarking. Finally, this thesis provides lessons learned and recommendations, and it aims to provoke thought among medical logisticians on the use of benchmarking.

#### B. RESEARCH METHOD

This thesis is based on a single case study. Robert Yin (1989) defines a case study as, "An empirical inquiry that investigates a contemporary phenomenon within its reallife context; when the boundaries between phenomenon and context are not clearly evident; and in which multiple sources of evidence are used." (p. 23) Case study research is appropriate when the research demands: an assessment of current events that were not influenced by the researcher; a variety of data sources are available; and answers to specific theoretical propositions are essential (Yin, 1989).

The use of the case study strategy is appropriate for this research for three reasons. The first criterion requires the assessment of a current event. The evaluation of the benchmarking study performed by the Materials Management Department at NMC San Diego constitutes the

assessment of a current event. Because people and documents are readily available to support the empirical research the second criterion is met. Finally, the third criterion requires answers to specific research questions. The answers to the research questions are gained by way of the research process by that telling what the researcher is studying.

#### 1. Data Gathering

Yin (1989) says, "Evidence for case studies may come from six sources of data: documentation, archival records, interviews, direct observation, participant-observation, and physical artifacts." (p. 85) For this research, literature reviews, interviews, reviews of documents, and a site observation were data sources for the case study.

#### a. Literature Review

A literature review was conducted to determine the commonly held themes regarding benchmarking and to identify: (1) a benchmarking model to use to describe and analyze the MMD experience, and (2) critical success factors necessary for an organization to address when attempting a benchmarking study. Over 32 pieces of literature pertaining to benchmarking were reviewed. Several books and periodicals were obtained by conducting a keyword search of the Naval Postgraduate School's BOSUN, a computerized card catalog. Additionally, the researcher consulted PROQUEST, an electronic journal database to locate articles on benchmarking. The researcher also made use of additional reference material available through the state of California inter-library loan service.

#### b. Site Visit

Due to time and monetary constraints only one site visit was conducted. Early discussions with the head of the Materials Management Department and Total Quality Leadership

Director at the Naval Medical Center San Diego deemed structured interviews would be the most effective method of data gathering. During the three day site visit, structured interviews, observations, and document reviews were conducted.

#### c. Interviews

Yin asserts, "[Interviews] can provide shortcuts to the prior history of the situation, so that the investigator can readily identify other relevant sources of evidence." (p. 91) The researcher used the results of the literature review on benchmarking to develop a list of 35 questions. Interviewee demographics such as: grade or rank, gender, title, and position held in the department were collected. The remaining questions focused on the individual's understanding of how MMD conducted its benchmarking study. Particular emphasis was placed on gaining information about why the benchmarking study was conducted, study participants, how the department planned the study, how the study was actually conducted, levels of leadership support provided, use of outside assistance, training conducted, lessons learned, rewards and incentives given, and finally publicizing of study results. The appendix contains the interview form and exact questions used.

Initially, interviewees were selected based upon their direct involvement in the benchmarking study. At the time of the interview a confounding factor emerged: many of the original study participants either had changed duty stations or had retired. This forced the researcher to relax the criteria of direct involvement in the benchmarking study to permit interviews with individuals who were

indirectly related to the benchmarking effort. This resulted in seven people being interviewed. They included:

- The department head who sanctioned the study and tasked the division officer with conducting the study.
  - The assistant department head of Materials Management who was not present during the actual study. However, he was given the assignment of overseeing implementation of the results of the study.
  - The assistant department head of Material Operations who arrived during the final stages of the benchmarking study and served as the key administrator of the Sterile Processing Division, the division where the benchmarking study was focused.
  - The division officer of Sterile Processing who was responsible for the day-to-day operations of the process.
  - The supervisor of Sterile Processing who manages the personnel engaged in the new process.
  - The technician of Sterile Processing who was peripherally involved in the study.
  - The Program Director of the Office of Continuous Improvement who had involvement in the command's development of their mission, vision, and strategic plan.

Interview questions were faxed to the assistant department head for distribution before arrival of the researcher. Interviews lasted approximately one (1) hour and the questions listed in the appendix were asked of each of the seven (7) interviewees. Participants answered all of the questions except the supervisor of the Sterile Processing Division and the technician within the division.

Both were not able to answer some questions because of time on the job and level of involvement in study. During each session notes were taken and a tape recorder was used to fill-in gaps of written notes. All participants spoke candidly about their experience with the benchmarking study. At the conclusion of each interview, participants were asked if they were interested in the findings of the study, and four of the seven participants expressed interest.

The researcher conducted an informal interview with an expert in logistics and benchmarking. This was done to gain a frame of reference for the specific area of logistics benchmarking.

#### d. Document Reviews

At the time of the site visit, the command's strategic plan, training records and award citations were reviewed to support specific responses given by interviewees during interview sessions. The command's strategic plan was used because it provides the framework within which the organization operates. It delineates why the organization exists, its purpose, when and how it intends to accomplish its objectives. Training records were used because they revealed who had received related training in benchmarking. Finally, award citations were used to determine the type of rewards and incentives given to the benchmarking team by command leadership.

#### e. Observations

The observations ranged from evaluating the climate of the department to considering the physical office environment. The evaluation of the climate of the department was performed to uncover factors pertaining to quality of interpersonal dynamics, readiness for change, and quality of communication. The physical environment was

considered because it provided insights on how benchmarking information was publicized in the work spaces.

#### 2. Analytic Strategy

This section describes how conclusions from literature reviews, interviews, document reviews, and a site observation were used to answer the primary and secondary research questions outlined in Chapter I of this thesis. A content analysis of the benchmarking literature was conducted to: (1) gain a consensus from the leading authors of the critical success factors for benchmarking, and (2) identify the most appropriate model for describing and analyzing the case study. Content analysis of the interview transcripts, notes, and documentation served as the foundation for writing the benchmarking case. Interview content analysis uncovered hindering factors, positive factors, and lessons learned. Finally, a comparison of the literature to the MMD experience was conducted to evaluate the success of the MMD benchmarking study.

#### III. LITERATURE REVIEW

This chapter reviews the benchmarking literature and focuses on determining the critical success factors for benchmarking. The chapter begins by defining benchmarking and describing reasons for benchmarking. The chapter concludes by examining the critical success factors for benchmarking and identifying the resources required to perform a benchmarking study.

#### A. BENCHMARKING DEFINED

In the late 1970's, Xerox Corporation had experienced major losses of market share to foreign competition. Afterwards, the corporation began an ambitious total quality management program to regain those market shares. The corporation used benchmarking to reclaim its competitive edge in the face of intense global competition. Xerox's Robert C. Camp pioneered the technique and defined it as, "The search for industry best practices that lead to superior performance." (Camp, 1989, p. 12)

Today, definitions of benchmarking abound. The American Productivity and Quality Center (APQC) defines it as, "The practice of being humble enough to admit that someone else is better at something and wise enough to try to learn how to match and even surpass them at it." (APQC, 1994, p. A-5) A practitioner defines it as, "A continuous search for and application of *significantly better practices* that leads to superior competitive performance. [italics added]." (Watson, 1993, p. 2) An expert described benchmarking as, "The process of understanding your performance, comparing it against the performance of bestin-class companies, learning how they perform better, and using that information to improve." (George, 1992, p. 75)

To this end, benchmarking can be summarized as a quality management tool used by organizations to accelerate change and maximize their competitive advantage through identifying and applying best practices.

While understanding what benchmarking is is vital, knowing what it is not is just as significant. Often organizations misunderstand benchmarking and benchmarking studies fail because people use the technique in situations that are incompatible with its original intent. Camp (1989) insists that benchmarking is not, "A mechanism for determining resource reductions although resources are redeployed in the most effective way of supporting customer requirements and obtaining customer satisfaction as a result of benchmarking activities." (p.14) Other authors profess benchmarking is not imitating, collecting statistics, visiting sites, or keeping to a standard by which something can be measured or judged, nor is it limited to a specific industry (DeToro, 1995; Grayson, 1994; Sheridan, 1993; Tutcher, 1994; Vasilash, 1994). It is not a cure all for all the ills existing in an organization.

This study defines benchmarking as,

The process of understanding your performance, comparing it against the performance of best-inclass companies, learning how they perform better, and using that information to improve. (George, 1992, p.75)

This definition is comprehensive, understandable, and relevant. It provides the researcher focus for the content analysis of the literature, assessment of interview responses, and derivation of lessons learned during the benchmarking study performed in the Materials Management Department at NMC San Diego.

Finally, before concluding this section it is important to note that the literature includes four types of benchmarking studies:

- Internal. An organization compares its own process with that of a similar or dissimilar process within the same organization(DON TQL Office, 1996).
- Competitive. An organization compares itself against its rivals in the same industry (Camp, 1989; DON TQL Office, 1996).
- Functional. An organization compares a job or function (i.e. marketing, warehousing, records keeping, etc.) to another organization in the same or different industry (Camp, 1989; DON TQL Office, 1996).
- Generic. An organization looks at the same function in similar or dissimilar industry (Camp, 1989; DON TQL Office, 1996).

#### B. REASONS FOR BENCHMARKING

Organizations initiate benchmarking studies for a variety of reasons. According to the literature they use it to:

Set Goals and Objectives. Benchmarking forces an organization to move from an internal improvement focus to looking externally for improvement goals. This external comparison gives the organization a broader perspective of the effectiveness and efficiency of their processes as compared to "best-in-class" organizations. For example, a group of individuals in one firm meets annually with a group of individuals from another firm at a "leadership summit" to compare business strategies and exchange business ideas to maintain a competitive edge in their industry (Fisher & Larsen, 1996). Motorola, another example, encourages its suppliers to look internally then look to others for ways to continually set new standards to improve their performance in the marketplace (Spendolini, 1992). This process helps Motorola and their suppliers because a continuous process improvement loop is established that relates to how to set goals and objectives (Spendolini, 1992).

- Enhance Continual Improvement. Benchmarking encourages an organization to continually learn from others to improve procedures, methods, and plans. In Eastern Technologies' case they fine tuned their setup procedures for new accounts by serving as a repository for firms to suggest new ways of doing business (Beasley & Cook, 1995).
- Improve Customer Satisfaction. Benchmarking can improve customer service by continually and systematically analyzing customer opinion. For example, organizations in the hospitality industry can use a combination of, "customer surveys, focus groups of previous and potential customers, use of fictitious guests, operational analysis of performance, employee circles, and employment of seasoned consultants" to analyze customer patterns. Firms then benchmark this data against their competitors in the marketplace (Yasin & Zimmerer, 1995).
- Understand World-class Performance.

Benchmarking encourages firms to seek out bestin-class organizations and work to emulate applicable processes. One electronic control manufacturer confronted with increased competition, dwindling profits and decreasing market shares looked to world-class performers to find solutions to improve the quality of their product parts and replacement components (Ventucci, 1992).

Whatever the motivating factor(s), an organization should recognize that benchmarking is most notable for improving processes to gain a competitive advantage. The decision to benchmark should not be based upon some shortterm crisis but to gain a long-term competitive edge.

#### C. THE BENCHMARKING PROCESSES

An important part of benchmarking is selecting a benchmarking process that will lead to a timely, practical, and a successful outcome. An organization should answer two questions to decide which model best serves their purpose. First, does the process take a system's view of benchmarking? Second, are the steps in the model comprehensive? Organizations are "Open systems that are influenced by a multitude of environmental forces or inputs such as availability of raw materials, changes in technology, competition, changing worker values, governmental policies, and so forth." (Bowditch & Buono, 1990) Open systems "interact with their environment and are influenced by external forces." (Bowditch & Buono, 1990) Thus, for a benchmarking model to have a systems view, it must take into account the entire organization. This includes the human, structural, and mechanical dimensions and the environment within which the organization exists.

A benchmarking model is comprehensive when it addresses in detail all steps required to conduct a successful benchmarking process and is easily understood by the benchmarking team and members of an organization.

Keeping in mind the requirement for a system's view and comprehensiveness, three models were analyzed: a Four-Step Process Model, a Nine-Step Quality Improvement Process, and a Ten-Step Benchmarking Model.

#### 1. Four-Step Benchmarking Model

Figure 3.1 illustrates the American Productivity and Quality Center (APQC) Four-Step Benchmarking Model. This

model, along with four others is presented during a two-day course sponsored by the APQC (APQC, 1994).



#### Step 1. Plan

Senior leadership forms a benchmarking team. The team's membership should consist of stakeholders of the process. Stakeholders are individuals and groups inside or outside the organization who affect and are affected by an organization's, group's or processes' mission, goals, and strategies. The team selects the process that is benchmarked. They define the scope of the study, determine the type of benchmarking effort, flowchart the internal process, and identify partners to benchmark (APQC, 1994).

#### Step 2. Collect Data

The benchmarking team determines methods of data collection (i.e. telephone or mail surveys, site visits, telephone interviews, face-to-face interviews, etc.). Data is collected from internal and external sources, and it is then matched (APQC, 1994).

#### Step 3. Analyze

The benchmarking team analyzes the data and determines performance gaps. The team assesses practices, methods, and procedures of best-inclass (APQC, 1994).

#### Step 4. Adapt and Improve

The benchmarking team communicates findings to senior leadership for acceptance. Once accepted the team develops a plan of action to implement the results and consistently recalibrates its findings (APQC, 1994).

The APQC Four-Step Benchmarking Model is a good model for organizations that are just starting their benchmarking effort. Remembering it is easy and is applicable to most benchmarking studies. One shortcoming of this model is that it may not meet the need of some government organizations to have detailed instructions for each phase of the model. Another drawback is that the model gives little attention to a system's approach to benchmarking. Otherwise, this is a reliable model.

#### 2. Xerox Corporation Nine-Step Quality Process

The Nine-Step Quality Improvement Process was developed by Xerox Corporation (Figure 3.2). Essentially, Xerox benchmarks at every step of the Quality Improvement Process, always comparing their results with competitors or best-inclass companies for that particular process (Xerox, 1993).



#### Step 1. Identify Output

The individual or group determines the product desired and then compares their findings against their competitor's product (Xerox, 1993).

#### Step 2. Identify Customer

The individual or group determines who the customer is for the process and then compares findings against their competitor's data (Xerox, 1993).

#### Step 3. Identify Customer Requirements

The individual or group determines the needs of the consumers of the product and then compare findings against their competitor's data (Xerox, 1993).

## Step 4. Translate Requirements into Supplier Specifications

The individual or group communicates the needs of consumers to suppliers and then compares results against their competitor's data (Xerox, 1993).

#### Step 5. Identify Steps in Work Process

The individual or group determines what tasks need to be performed before a product is placed on the market and then compares the process against their competitor's process (Xerox, 1993).

#### Step 6. Select Measurements

The individual or group identifies how outcomes are evaluated and then compares the process against their competitor's outcome data (Xerox, 1993).

#### Step 7. Determine Process Capability

The individual or group identifies strengths and weaknesses of current procedures and then compares the process against their competitor's procedures (Xerox, 1993).

#### Step 8. Evaluate Results

The individual or group assesses outcomes of processes and then compares the results against their competitor's data (Xerox, 1993).

#### Step 9. Recycle

The individual or group makes improvements where necessary and then compares the results against their competitor's data (Xerox, 1993).

This nine-step quality improvement process is excellent for an organization that is more experienced in benchmarking. This is a mature benchmarking model in that the preparatory steps generally outlined in other models are not detailed in this model. Another unique characteristic of this model is that it is not limited to a group conducting a benchmarking study; an individual can use the process. Overall, the process is most appropriate to offer an organization when it has quality immersed in its culture.

#### 3. Ten-Step Department of the Navy (DON) Benchmarking Model

The final model analyzed was the Ten-Step DON Benchmarking Model as shown in Figure 3.3. The Office of the Under Secretary of the Navy Total Quality Leadership Office drafted this benchmarking model, which consists of four basic phases: Plan, Do, Study, Act. This model is similar to the Plan-Do-Check-Act Cycle: A Method for Continual Improvement taught in Total Quality Leadership training courses (Fundamentals of Total Quality Leadership, 1992).

#### Plan Phase

The Plan Phase of the DON Benchmarking Model begins with identifying operations that need improvement. In this phase a benchmarking team identifies, "what to benchmark, benchmarking partners from best-in-class and determines data collection methods." (DON TQL Office, 1996)

#### Do Phase

The Do phase consists of, "comparing internal versus external performance gaps." (DON TQL Office, 1996) In this phase a benchmarking team decides, "performance gaps and strengths and takes a system's view." (DON TQL Office, 1996)

#### Study Phase

The Study Phase involves assessing the, "acceptance of benchmarking." (DON TQL Office, 1996) In this phase a benchmarking team, "communicates benchmarking findings and gains acceptance, and supports and establishes functional goals." (DON TQL Office, 1996)

#### Act Phase

The Act Phase is the final phase in the process. It involves, "developing a method to carry out findings and to recalibrate when appropriate." (DON TQL Office, 1996) In this phase a . benchmarking team, "develops an action plan, implements best practice procedures and monitors progress then recalibrates benchmarks." (DON TQL Office, 1996)



#### Plan Phase

#### Step 1. Identify What to Benchmark

The Executive Steering Committee (ESC) identifies potential processes to benchmark. They identify a process that aligns with the organization's

strategic plan. They define the scope of the study, determine the benchmarking effort, and flowchart the broad functional process. The ESC then charters a benchmarking Quality Management Board (QMB) to serve as the champion of the potential process to benchmark. The QMB charters a Process Action Team (PAT) to carry out the study (DON TQL Office, 1996).

#### Step 2. Identify Benchmarking Partners from Bestin-Class

The PAT flowcharts the internal process. They then review literature, interview industry experts, and more to decide who is best-in-class. Afterwards they consider appropriate methods for data collection. Finally, the PAT flowcharts the process of the best-in-class (DON TQL Office, 1996).

## Step 3. Determine Data Collection Methods & Collect Data

The PAT selects a method for data collection and then collects data from external sources. Once the data is collected the team match up common processes, procedures, or methods between the internal and external operations of each organization (DON TQL Office, 1996).

#### Do Phase

#### Step 4. Determine Performance Gaps and Strengths

The PAT evaluates the organization's practices in the process under study against that of industry's best to decide which practices, methods, or procedures should be changed. The PAT concludes its study by submitting a report of findings to the QMB (DON TQL Office, 1996).

#### Step 5. Take a System's View

The QMB evaluates the report from the PAT and determines impact on entire organization. The QMB makes recommendations to the ESC. The ESC reviews recommendations and analyzes the impact on the
organization's mission, vision, and strategies (DON TQL Office, 1996).

Study Phase

# Step 6. Communicate Benchmarking Findings and Gain Acceptance and Support

The results of the study are told to the appropriate levels of the organization. Acceptance and support are sought from all appropriate levels of the organization (DON TQL Office, 1996).

#### Step 7. Establish Functional Goals

The ESC and QMB evaluate current jobs to decide if revisions should occur to align the process with accomplishing the command's strategic plan. They obtain a commitment from senior leadership to implement findings (DON TQL Office, 1996).

#### Act Phase

#### Step 8. Develop Action Plan

The QMB and PAT draft a plan of action for the implementation of the results. During this step they develop mechanisms to measure and monitor achievement of the benchmarking results (DON TQL Office, 1996).

## Step 9. Implement Best Practices Procedures and Monitor Progress

The ESC approves the plan of action and allows process owners to implement changes (DON TQL Office, 1996).

#### Step 10. Recalibrate Benchmarks

At this step it is up to all levels of the organization to monitor the process and continuously improve upon it (DON TQL Office, 1996).

This model is most appropriate for organizations considering benchmarking for the first time or organizations that are experienced bench markers. It takes a system's view to benchmarking and is comprehensive. For these reasons, the researcher selected the Ten-Step DON Benchmarking Model to evaluate the success of the benchmarking study and develop lessons learned from the process.

# D. CRITICAL SUCCESS FACTORS FOR BENCHMARKING

This section discusses the critical success factors for successful benchmarking. Critical success factors are "those characteristics, conditions, or variables that have direct influence on customers' satisfaction with the output--product or service or both--of specific business processes and hence, are critical to the success of the entire business." (Watson, 1993, p.57) The following themes represent critical success factors for benchmarking:

## 1. Understand Your Own Process First

Experts in the field agree that firms gain the greatest insight about themselves when they conduct benchmarking studies (Petrick, Scherer, Watson, 1993; Westfall, & Wilson, 1994; Venetucci, 1992). As Eric Kennedy puts it, "You must have your own house in order." (Human Resource Planning, 1993) When organizations attempt to understand and document the process under study, they learn why they do the things they do and how these actions affect the external environment (Tutcher, 1994). For example, in the beginning of the AT&T Oklahoma City Works benchmarking experience, the team met regularly to: (1) discuss their benchmarking effort, (2) create measures of effectiveness, and (3) map current practices (Pulat, 1994). This initial work lead to more benchmarking studies conducted at AT&T and more

employees having a clear understanding of benchmarking (Pulat, 1994).

# 2. Make Benchmarking a Part of Strategic Planning

Several authors assert that making benchmarking a part of strategic planning fosters a philosophy of continuous process improvement throughout the organization (DeToro, 1995; Greengard, 1995; Tutcher, 1994; Watson, 1993). Organizations have two choices when integrating benchmarking with strategic planning. They can state it explicitly as a part of an organization strategy or they can develop goals and objectives that require the use of benchmarking to accomplish the mission and vision of the organization. For example, in the 1994 draft of the Naval Medical Logistics Strategic Plan, leaders in the medical logistics community stated "Benchmarking shall occur with other organizations." (Strategic Plan, 1994) By contrast, when Caterpillar, Inc. went through major restructuring, benchmarking emerged as a tool for implementing the strategic plans of their Technical Services Division (Mittelstaedt, 1992).

#### 3. Leadership Support and Commitment

Benchmarking requires extensive leadership from senior managers in the organization, as for demonstrating commitment to the process, and, most important, providing the financial backing necessary to conduct the studies.

Many authors agree that without leadership support and commitment benchmarking teams could be given inadequate attention and resources (DeToro, 1995; Mittlstaedt, 1992; Tutcher, 1994; Watson, 1993). Becoming a champion is an ideal way to show commitment to the benchmarking process. A champion means "One that defends, fights for, or supports a cause or another person." (The American Heritage Dictionary, 1991, p.257) Robert Camp is a living example of

a champion. He is known around the world for his contributions to benchmarking at Xerox Corporation (Etorre, 1993). It is important for a champion to regard benchmarking as a viable tool to propel the organization to a higher plateau. Champions should be "encouraged from the top of the organization to pursue the study, and their results should be communicated to encourage others who need benchmarking. . ." (Watson, 1993, p.197).

## 4. Education and Training

Education and training in benchmarking process, tools, and techniques are critical to the success of the benchmarking effort (Mittelstaedt, 1992; Sillyman, 1992; Spendolini, 1992). It helps employees in understanding the process and transforming organizations. One author recommends that participants of benchmarking studies read books and articles about the topic plus attend conferences (Mittlesteadt, 1992). Xerox understood the importance of education and training. The leadership wanted company employees to embrace the "quality effort" so everyone in the organization was given training in quality skills and tools (Xerox, 1993). At Goodyear Tire and Rubber Co., employees received just-in-time benchmarking training when they are prepared to measure themselves against their competitors (Greengard, 1995).

## 5. Communication is Paramount

The entire literature emphasizes the importance of communication through all levels of the organization. Experts agree that when a people share a common set of objectives and an agreed upon view of the world, coordinated and focused action is much easier to accomplish (Camp, 1989; Leibfried and McNair, 1992).

Watson (1993) emphasizes the need to share benchmarking stories. When benchmarking stories are shared people are aware of the benefits and the organization has a greater chance at performing a successful benchmarking study (Watson, 1993). Nationally, AT&T bench markers share their stories on-line (Pulat, 1994). Specifically, AT&T Oklahoma City Works conducts "sharing forums" regularly to discuss benchmarking experiences (Pulat, 1994).

## 6. Provide Adequate Resources

The literature states that benchmarking requires a considerable commitment of time, talent and other resources.

## a. Time Commitment

Two sources asserted that managers frequently underestimate the time required to perform a benchmarking study (DeToro, 1995; Sheridan, 1993). Ironically, consultants seem just as quilty. In the case of a prominent firm, a benchmarking team was given four weeks to plan, collect, analyze, adapt and improve a process (Sheridan, 1993). This was an unusual case; rarely can any benchmarking study be completed in four weeks. One consultant exclaimed that it took six months, several experts, and numerous outside agencies to complete one benchmarking project (Ettorre, 1993). A representative from the AT&T Benchmarking Group had this say about time commitment: "It takes between four and six months for an average benchmarking project, maybe longer if you do not have much experience at it." (Mittelstaedt, 1992) One Malcolm Baldridge Award Winner claimed employees dedicated on average of five hours out of a 40-hour work week to their benchmarking effort (Godfrey, 1995). Overall, The American Productivity and Quality Center estimates that benchmarking teams meet one day per week and that studies be completed

within six months, although they can last from three to twelve months (APQC, 1994).

#### b. Proper Team Composition

Organizations should select people who represent different operations within an organization for the benchmarking team because they are the ones who will know the process and are able to correct problems (DeToro, 1995). The American Productivity and Quality Center recommends assigning individuals who have a genuine interest in either benchmarking or total quality management (APQC, 1994). Also, the center advocates creating benchmarking teams of five to seven people who can meet weekly (APQC, 1994). When Kodak's Human Resources Department wanted to do a benchmarking study, they not only selected individuals who were stakeholders of the process but also individuals who were not familiar with the process; these individuals could bring a different perspective (Greengard, 1995). Generally, a smaller group is easier to manage. The people assigned to benchmarking teams should have a personal stake in the process. Making team membership mandatory or selecting individuals for the team that are not interested in benchmarking is counter productive to a successful benchmarking study.

# c. Cost

Finally, organizations that have completed benchmarking projects report that they have spent between \$25,000 and \$150,000 (Buckler, 1994). An analysis conducted by the American Productivity & Quality Center echoed these findings by revealing that the average cost to complete a study ranges from \$35,000 to \$70,000 (APQC, 1994). The money spent on benchmarking studies is allocated to items such as: team training, searches, site visits, implementation costs,

membership fees, annual fees, and initiation fees. For example, if an organization were seeking membership into the International Benchmarking Clearinghouse, they would pay an annual membership fee of \$6,000 along with a one-time initiation fee that could be as much as \$12,500 (the amount depends on the number of employees in the organization (APQC,1994).

Overall, the time, talent, and other resources required to complete a benchmarking project is dependent upon the magnitude of the study and the number of employees in the organization (Buckler, 1994). However, the savings from this kind of investment is invaluable. One organization reported saving between \$5-\$6 million annually as a result of performing a benchmarking study (Buckler, 1994). In the case of Xerox, they saw remarkable savings in labor and productivity. The rejection rate of defective machines was reduced too less than ten percent. The distribution productivity went from 5 percent to 10 percent, and a dramatic decrease in service labor cost occurred (Mittelstaedt, 1992).

## E. SUMMARY

This chapter began with discussing what benchmarking is and what it is not and reasons for benchmarking. While understanding that benchmarking is a useful tool for managers to improve customer satisfaction knowing that benchmarking cannot cure all the ills of an organization is important. Benchmarking reasons stem from its ability to help organizations assess their internal processes before looking externally for improvement goals. The area an organization is interested in improving can lead to a benchmarking study being done.

The remaining section discussed the benchmarking model used to evaluate the benchmarking study and the critical success factors for benchmarking. Three benchmarking models were analyzed: a Four-Step Process Model, a Nine-Step Quality Improvement Process, and a Ten-Step Benchmarking Model. The Ten-Step Department of the Navy Benchmarking Model was selected because it met the researcher's requirement for a system's view and comprehensiveness. This model is designed for new or seasoned bench markers. The model offers a great deal of detail and guidance for bench markers to do a reliable benchmarking study.

As previously mentioned, the critical success factors for benchmarking were discussed. These essential characteristics should exist before a benchmarking study is initiated in an organization. The most common factors that emerged from the literature were understanding your own process first, benchmarking should be a part of strategic planning, leadership support and commitment are important, education and training are necessary, communication is paramount, and adequate resources are required. The importance of each critical success factor should not be underestimated because together they represent a model framework for benchmarking success.

#### IV. THE CASE STUDY

This chapter presents the study conducted in the Materials Management Department (MMD) at the Naval Medical Center (NMC) San Diego. The first section describes the study. The final section summarizes responses from the interview questionnaire.

#### A. BACKGROUND

The study was conducted in the Sterile Processing Division within the Materials Management Department of the Naval Medical Center San Diego. The Sterile Processing Division is responsible for sterilizing and wrapping instrument sets for surgery. The division sanitizes surgical instruments with Ethylene Oxide (EtO) sterilizers that use chlorofluorocarbons (CFCs). Ethylene oxide is,

a colorless, flammable toxic gaseous or liquid compound  $C_2H_4$  made by reaction of ethylene chlorohydrin and alkali or by catalytic oxidation of ethylene and used chiefly in organic synthesis (as of ethylene glycol and ethanolamine) and in sterilization and fumigation. (Webster's Third New International Dictionary, p.781)

### A chloroflurocarbon (CFC) is,

any various halocarbon compounds of carbon, hydrogen, chlorine, and fluorine, once used widely as aerosol propellants and refrigerants. Chlorofluorocarbons are believed to cause depletion of the atmospheric ozone layer. (The American Heritage Dictionary of the English Language, 3rd, p. 336)

Both are harmful to your health. Scientific research indicates that CFCs thin the ozone layer and ethylene oxide pollutes the air (L.A. Times, 1994).

The United States Congress and local governments have taken steps to minimize the harmful effects that these substances have on humans, crops, and marine life by promulgating legislation. For example, the Congress has imposed a hefty excise tax on the carrier gas CFC in hopes of discouraging future use and production. Also, several state governments now require manufacturers to find alternate ways to minimize ethylene pollution.

The medical industry relies heavily upon a mixture of 12% chlorofluorocarbon (CFC-12) and 88% ethylene oxide (EtO) to sterilize medical instruments (OR Manager, 1994). The new mandates have forced many medical facilities, including the Materials Management Department at NMC San Diego, to find alternative methods of sterilization. The following paragraphs describe the benchmarking study conducted to determine the most ecological method of sterilization.

### B. THE STUDY

#### 1. Planning Phase

#### a. Identify What to Benchmark

In March of 1995 the Sterile Processing Division began studying their sterilization process. The head of the Materials Management wanted to: (1) comply with federal regulations, (2) minimize employee exposure to known carcinogens, and (3) eliminate equipment that polluted the air. The department head recommended the Sterile Processing Division conduct a benchmarking study to determine the best method of sterilization. He had heard of benchmarking during professional conferences and strategic planning efforts and was familiar with its benefits. Shortly after his recommendation a team was formed. The team consisted of four people:

- The assistant department head of Materials Operations;
- The division officer of Sterile Processing;
- The supervisor in the Sterile Processing Division; and
- The operating room nurse.

Initially, study team members were not assigned specific roles (i.e. Executive Champion, Benchmarking Manager, Research Manager, Information Specialist, or Project Facilitator) to perform during the study. Further along in the process, the assistant department head of Materials Operations became the team leader. The department's training report indicates that most of the members on the team had little experience with benchmarking, little training in total quality leadership, and minimal exposure to others who had conducted benchmarking studies. The interviewees were not sure if any of the team members ever had seen a successful benchmarking study. There was no evidence of the team being given a charter by a Quality Management Board that explained their scope, delineated possible time lines, identified the type of benchmarking effort, listed critical success factors, or specified desired outcomes for their study. Therefore, the team appeared to have little guidance from higher authority and less knowledge of the overall impact of their study on the command's strategic plan. According to interviewees, the team gave little attention to using benchmarking models or flow charts to explain their processes. The department head was available to answer questions if the team encountered a problem.

The interviewees were asked general questions regarding the formulation of the team. Three of the seven interviewees were not aware of the exact composition of the team, for one of the following reasons: (1) the interviewee was peripherally involved in the study, (2) the interviewee was not present during the study, and finally (3) the Director of the Continuous Improvement Office was not aware of the study being conducted. The research indicated that there were five key players during the study, and they were responsible for the following:

- The assistant department head of Materials Operations led the initial phase of the study. By default, he became the leader of the study because he was the administrator over the Sterile Processing Division. Later, the assistant department head was replaced after receiving orders to a new duty station.
- The division officer of Sterile Processing served as the manager who was most familiar with the sterilization process.
- The supervisor of Sterile Processing served as the resident expert on sterilization.
- The operating room nurse represented the surgery department who was a major user of sterilized surgical instruments. This individual also possessed the knowledge about plasma sterilizers.

Based on the interview data, it appears that no formal appointments were made to this team. Additionally, the data did not show a cross representation of customers or involvement of other members in the Materials Management Department (e.g. enlisted personnel or civilians).

With the exception of one interviewee, the informants agreed that the motivating factors behind the

study were not related to a fear of a base closure, visionary leadership, competition, or downsizing. The primary reasons for the study were to comply with federal regulations, to minimize known human carcinogens, and to eliminate equipment that polluted the air. During the planning phase of this study, a limited number of manufacturers were identified for best-in-class. The selection of the manufacturers was based upon the prior. knowledge of a team member. There was no evidence that a formal plan of action was developed.

Most of the interviewees were unable to give a precise account of all the steps that occurred when the team conducted the study. For the most part, the informants did not recall the use of a benchmarking model or specific reference materials. The interviewees were comfortable estimating that the study took less than a year to complete and they felt that the time devoted to the study was reasonable for the task assigned.

# b. Identify Benchmarking Partners from Best-in-Class

Before identifying partners from best-in-class the team needed to answer three critical questions: (1) should they retrofit the current system with hydrochloroflourocarbons vice chloroflourocarbons, (2) should they purchase a 100% EtO Sterilizer, or (3) should they completely replace the existing system with a plasma sterilizer? The type of study conducted would be based upon answers to the above questions. A cost analysis indicated that retrofitting or purchasing new EtO Sterilizers would be cost prohibitive. Additionally, the use of EtO Sterilizers would require compliance with strict air pollution regulations. The team concluded that the best alternative was to completely

replace the existing system with a plasma sterilizer. The team then identified Johnson & Johnson and Abtox as two manufacturers of plasma sterilizers. The two manufacturers were selected based upon the operating room nurse's prior knowledge of firms that manufactured plasma sterilizers.

# c. Determine Data Collection Methods & Collect Data

The team used newspapers, articles written in professional journals, and telephone interviews to collect data related to these two manufacturers of plasma sterilizers. This method of data collection was selected because a majority of the information was readily available in the medical library of the Naval Medical Center San Diego. The data then was matched in the following areas: (1) processing time, (2) limitations of system, (3) advantages of system, (4) approval or disapproval status of the system by the Food & Drug Administration (FDA), and (5) application of the system to heat-sensitive equipment.

#### 2. Do Phase

#### a. Determine Performance Gaps

It was difficult to measure performance gaps in this study because the study was not focused on comparing the MMD sterilization process to best-in-class processes at other medical facilities.

#### b. Take a System's View

It was clear that the team was concerned with minimizing employee exposure to and elimination of equipment that caused pollution; however, according to interviewees, the driving force behind the study was to ensure compliance with federal regulations. The team concluded that the plasma sterilizer manufactured by Johnson & Johnson offered the best features needed by NMC San Diego. To the best of

the knowledge of the interviewees, these findings from the study were not formally presented to a senior management board for review or comment.

## 3. Study Phase

# a. Communicate Benchmarking Findings and Gain Acceptance and Support

The team prepared a report of their findings and submitted the information to the department head. The department head was pleased with the findings and approved the purchase of the plasma sterilizer manufactured by Johnson & Johnson. The news regarding the purchase of the plasma sterilizer was shared with members of the medical staff, operating room nurses, and technicians who depended on sterilized medical instruments to perform their jobs. According to interviewees, each group welcomed the new technology and regarded it as a plus for the environment and employee exposure to hazardous substances.

## b. Functional Goals

To the best of the knowledge of the interviewees, new functional goals were not incorporated into the command's strategic plan by the ESC. The results were implemented at the department level by department personnel. Neither the ESC nor any QMB had the opportunity to witness implementation of any of the new practices discovered during the study because they had not sanctioned the study and were not aware of the study occurring.

## 4. Act Phase

## a. Develop Action Plan

Since the team did not develop a formal plan of action to implement the new sterilizer, it was difficult to track how the team intended to measure the results of the study. Basically, the plan of action centered around

informal verbal agreements between team members and the department head to initiate the procurement process of the new sterilizer. The department head had final approval of the purchase of the new equipment, according to interviewees.

# b. Implement Best Practice Procedures and Monitor Progress

After Johnson & Johnson was selected as the manufacturer of choice, a visit to the plant where the plasma sterilizer was manufactured took place. The team was convinced that the sterilizer by Johnson & Johnson was the best system to implement in the Sterile Processing Division. Acquisition of the system was initiated. The system arrived within 4-6 months after the initial paperwork was done. The system has self monitoring capabilities that measure the level of living bacteria and other microorganisms during the sterilization process, therefore little human interaction is required and measurement of original study goals is maintained.

## c. Recalibrate Benchmarks

By the end of the study, the team had accomplished what it had set out to do. The team satisfied the needs of the hospital by successfully finding a system that could comply with federal regulations, minimize employee exposure to known carcinogens, and eliminate equipment that polluted the air. The study neither lent itself to instituting formal procedures to recalibrate the benchmarking effort, nor did it put any standardized procedures in place to identify emerging world-class processes for sterilization.

## C. INTERVIEW RESPONSES

In addition to describing the study process, interviewees were asked a number of questions related to the

support the study received, lessons learned, and benefits of conducting the study. These responses are summarized using the categories of the interview questionnaire (Appendix).

#### 1. Benchmarking Defined

For this study, the researcher defined benchmarking as,

The process of understanding your performance, comparing it against the performance of best-inclass companies, learning how they perform better, and using that information to improve. (George, 1992, p.75)

One interviewee defined benchmarking as, "A reference point, a landmark." The researcher gathered that this definition was similar to the one given in the American Heritage Dictionary. Another respondent defined benchmarking as, "A process that you are doing now and compare it with other processes that are similar and you try to do a better job," while another defined it as, "Looking at what other facilities are doing and seeing how your organization can adopt those practices. Ultimately, you may or may not adopt the practices."

#### 2. Leadership Support

It was difficult to identify leadership support during the study. Perhaps, the best illustration of this point is the comment made by an interviewee. The interviewee remarked that he, "Was not aware of top level management support but did know that the team had the support of the department head." This was a regular sentiment among most of the interviewees except for the department head who had this to say, "Support was provided by command leadership via monetary funds to purchase the new equipment." It was clear from the interviews that there were different interpretations of what leadership support and commitment is

and is not. Other issues of concern to the researcher were, involvement of a champion in the study and the integration of benchmarking into the command's strategic planning process. To the best of the knowledge of the interviewees, a champion was not identified during the study. Also, most of the respondents could not recall if benchmarking was integrated into the command's strategic plan. The exception was the Director of the Office of Continuous Improvement who said, "No, benchmarking is not integrated into the command's strategic plan."

## 3. Other Organizational Expertise

"Nursing Services may have done some benchmarking," one respondent explained. Another respondent said, "The command does perform comparative data analysis in some clinical and administrative areas." For the most part, the interview sessions revealed that other areas in the hospital may conduct modified benchmarking efforts; however, these resources were not sought out at the time of the benchmarking study.

#### 4. TQL Office

According to the Director of the Office of Continuous Improvement and one interviewee, benchmarking was discussed in Total Quality Leadership training courses offered by the command. Unfortunately, the other interviewees were unable to give substantial information regarding Total Quality Leadership training courses offered by the command or discuss the assistance the team obtained from the TQL office at the time of the study.

## 5. Education and Training

There was not a consensus from interviewees on the level of education and training provided to the team. In general, interviewees could not recall what training the

team may or may not have received at the time of the study. Finally, a review of the department's training report was performed. Establishing the level of education on benchmarking for the team was difficult because the Staff Education and Training Department's Standard Personnel Management System reflected a roster that consisted of staff currently assigned to the command. This limited the researcher's ability to analyze the type of training the team had received. From a broader perspective, it still is important to note that of the 182 people assigned to the Materials Management Department only 25 had received training in the Introduction of Total Quality Leadership. Specifically, two of the six interviewees had attended this training. Further research revealed that, 4 out of the 182 people had training in the tools of Total Quality Leadership and none of the six interviewees from the Materials Management Department had received this training.

## 6. Lessons Learned

Deriving the lessons learned from this study was not difficult for the interviewees. They commented on the importance of having defined goals before a benchmarking study is conducted. Second, they advised that anyone considering a benchmarking study should take into account the needs of the organization, the benefits of benchmarking, and the competitive advantage benchmarking offers. At the conclusion of this question and answer period, one interviewee pointed out a major shortcoming of the study. They explained, "The federal regulation was driving the change in the organization and not the desire to be the best-in-class or improve one's competitive position."

## 7. Follow-up

According to most of the interviewees, they could only speculate that all of the results of the study were adopted by management and top leadership. This was not the case for the department head who was confident that all results of the study were adopted. In general, the interviewees recognized the benefits of benchmarking and would consider participating in future benchmarking studies.

# 8. Rewards and Incentives

Most of the interviewees were aware of the recognition the team members received for their contribution towards the study. The military members earned achievement medals and the civilian members earned cash awards for their contributions toward the completion of the project. The research revealed that the team members were given awards that were comparable in significance to other awards given at the command.

# 9. Public Relations

Without exception, the interviewees agreed that publicizing benchmarking success stories was not a part of the command culture. In spite of this, three interviewees did mention that a report of the study was prepared and submitted to a professional medical leadership conference.

## V. ANALYSIS AND CONCLUSION

This chapter analyzes the case data collected on the study conducted by the Sterile Processing Division in the Materials Management Department at NMC San Diego. It includes an assessment of the research questions delineated in Chapter I and an analysis of the Materials Management Department study against the framework of the critical success factors for a study discussed in Chapter III. It also identifies areas for future research.

# A. MATERIALS MANAGEMENT DEPARTMENT EXPERIENCE VERSUS INDUSTRY THOUGHT

## How does the Materials Management Department experience compare with industry thought regarding benchmarking?

To answer this question the case study was compared with the definition of benchmarking, the reasons for benchmarking, the Ten-Step Department of the Navy Benchmarking Model, and the critical success factors discussed in the literature review.

#### 1. Definition of Benchmarking

As discussed in Chapter III the definition of benchmarking used for this study was,

The process of understanding your performance, comparing it against the performance of best-inclass companies, learning how they perform better, and using that information to improve. (George, 1992, p.75)

Interviewees had slightly different definitions of benchmarking which highlighted some of the elements outlined above, but were not as comprehensive. Most understood that benchmarking involved comparison with others, but not everyone understood that benchmarking is a defined process, nor that comparisons are made against "best in class"

processes. There was little evidence to show that standard methods (i.e. gathering customer satisfaction data, flowcharting or maintaining process metrics) were used to understand current performance. Additionally, there was no evidence that efforts were made to identify "best-in-class" organizations for comparison; the study concentrated on comparing two types of sterilization equipment.

# 2. Reasons for Benchmarking

As noted in Chapter IV, the Materials Management Department decided to conduct a benchmarking study to:

- Comply with federal regulations,
- Minimize known human carcinogens, and
- Eliminate equipment that polluted the air

Although laudable, these reasons do not address what the literature says a organizations should consider before undertaking benchmarking. Specifically, the literature asserts that one reason benchmarking studies are conducted is to set goals and objectives for comparing organizational processes with "best-in-class" processes in other organizations. The goals and objectives noted above do not include a desire to compare the effectiveness and efficiency of the Materials Management Department's sterilization processes to "best-in-class" sterilization processes.

Additionally, although the Materials Management Department study did result in enhanced productivity (another reason noted in the literature), this was not a reason the study was undertaken.

Another reason organizations conduct benchmarking studies is to continually improve their procedures, methods and plans. Improvements were realized in the areas of

safety and regulatory compliance. However, there was nothing in the data to suggest an emphasis on <u>continual</u> improvement in these areas nor any emphasis on continual improvement in other areas of sterilization quality such as sterilizing surgical instruments quicker. These types of improvements were not reasons the study was undertaken nor were they a criteria for selecting a specific sterilization system.

Organizations also conduct benchmarking studies to enrich customer satisfaction by constantly analyzing customer feedback and comparing it to "best-in-class" organizations. Though the department head was pleased with the results of the study and the surgical staff welcomed the new technology, establishing feedback loops to continually monitor customer satisfaction against industry's best practices were not the motivators behind this effort.

Finally, organizations conduct benchmarking to understand and emulate world class performance. Once again, there was no data to suggest this was a reason for undertaking the Materials Management Department study.

# 3. The Ten-Step Department of the Navy Benchmarking Model

The following sections compare the Materials Management Department study to the Ten-Step Department of the Navy Benchmarking Model discussed in Chapter III. The analysis is organized into the four phases of the model (Plan, Do Study and Act).

#### a. Plan Phase

The model suggests that the ESC is involved in every phase of the benchmarking study. During the planning phase they identify the potential processes to benchmark (aligned with the command strategic plan) and charter

appropriate teams (QMBs and PATs). There was no evidence to show that the ESC was involved in any portion of the Materials Management Department study or that the study was linked to the strategic plan. Also, there was no evidence to show involvement by the ESC or that a charter had been given to the team.

Another portion of the Plan Phase is identifying "best-in-class" organizations to benchmark and collecting data from that organization. Though the team collected data on plasma sterilizers and selected a sterilizer that best met the needs of NMC San Diego, there was no evidence to show that the team had identified or collected data from world-class sterilization processes as part of their study.

The benchmarking team also flowcharts the process under study during this phase. As noted in the case, there was no data that showed any processes were documented in the sterilization processing division. Rather the study focused on an analysis of different types of processing equipment, which would normally only be part of a full benchmarking study.

## b. Do Phase

In this phase, the PAT or other benchmarking team evaluates the organization's process against that of the benchmarking partner to discover performance gaps and strengths and submits a report of findings to the chartering QMB for further action. As the Materials Management Department study was focused on evaluating types of equipment verses comparing the Materials Management Department's sterilization process to a "best-in-class" process, this evaluation of performance gaps was not accomplished. An important element of the Do phase is taking into account the impact of the study recommendations

on the entire organization. This systems analysis does not appear to have been accomplished, perhaps because the department did not believe the study would have broad ranging effects on other areas of the hospital.

#### c. Study Phase

During the Study Phase, the model suggests that the benchmarking team present its findings to all appropriate levels of the organization and gain acceptance of their findings. The data shows that this was accomplished in the Materials Management Department study. The results of the study were presented to the decisionmaking authority (the Materials Management Department Head) and shared with and welcomed by surgical staff, the main customers. In addition, the study was presented at an external medical leadership conference. Although the study was not presented to the ESC, this appears appropriate given that the area under study was solely owned by the Materials Management Department Head, and that all outcomes from the study were implemented at the department level by department personnel.

Another portion of this phase is to establish goals to align the process with the command's strategic plan. This was not accomplished, however, given the fact that the study was specific to a process owned by one department it does not appear to have created systemic problems.

## d. Act Phase

According to the model, during the Act Phase the QMB and PAT plan the implementation of the benchmarking results. The ESC approves this plan and the entire organization actively participates in the continued improvement of the new process. Although the team had a

verbal agreement with the department head to procure the new equipment, there were no documents that reflected a plan of action and milestones for the implementation of a new sterilization process. The team did accomplish what it had set out to do, which was comply with federal regulations, minimize known human carcinogens, and eliminate equipment that polluted the air. Other than meeting these goals, there was no evidence that procedures were put in place to continuously improve the process associated with the new equipment.

# 4. Critical Success Factors

#### a. Understand Your Own Process first

This critical success factor directly relates to the Plan Phase of the Ten-Step Department of the Navy Benchmarking Model. According to the literature, an organization should understand their internal processes before conducting a benchmarking study. As the analysis in section A.3.a indicates, a detailed understanding of the sterilization process was not undertaken. Although this did not affect the results of the study, it does lend credence to the idea that the study conducted was technically not a benchmarking study.

# b. Make Benchmarking an Integral Part of Strategic Planning

The literature suggests that benchmarking be made an integral part of strategic planning. The case shows no evidence that benchmarking was either explicitly or implicitly articulated in NMC San Diego's strategic plan. In spite of this, the Director of the Continuous Improvement Office did say, "The command is looking at developing a department profile whereby department heads could track how well they were managing resources in comparison to practices

of other departments in the hospital." (Goff, 1996) In the final analysis, the researcher concluded that the integration of benchmarking into strategic planning is not yet a high priority for this command.

#### c. Leadership Support and Commitment

Benchmarking efforts are successful when they receive the necessary support and commitment from senior management. This aspect of the case was difficult to assess because the study was not directed by senior leadership. The researcher learned that the team received support from the department head; similar support was not gained from senior leadership (i.e. the Commanding Officer, the Executive Steering Committee, or the Quality Management Board) The limited involvement of senior leadership did not appear to be a problem because the study was contained within the Materials Management Department. The Department Head of Materials Management acted as the senior leadership and provided resources (i.e. time and money) and empowered the team to explore alternatives. Finally, though the department head chartered the benchmarking team, there was no evidence that he or anyone else acted as a champion to encourage the benchmarking effort.

# d. Timely Education and Training is Important

The literature review indicates benchmarking education and training are key ingredients to an organization's benchmarking success. Interviews and training reports did not offer evidence that the team had received benchmarking training. There was some evidence that very general discussions on benchmarking were presented in some of the command's Total Quality Leadership courses. However, as noted in the case study, very few Materials Management Department personnel had taken advantage of these

courses. This lack of education and training in benchmarking may have contributed to the study being limited to an analysis of two types of sterilization equipment verses a full benchmarking study.

## e. Communication is Paramount

According to the literature, communication with all levels of the organization is an important step to gaining acceptance of the benchmarking findings and . informing senior leadership of the team's results. Though the team's findings were not communicated throughout the organization or with senior leadership, there is considerable evidence to conclude they were communicated to appropriate levels of the organization. Based on the Study Phase information the right people were informed.

## f. Provide Adequate Resources

When an organization decides to benchmark, they should consider the team composition, the cost of the study, and time commitment of employees towards the benchmarking effort. The study showed four people participating in the effort, which appeared to be an appropriate number for this type of study. The composition of the team was restricted to individuals directly affected by the new sterilization process. There was little evidence available about the cost of the study to the command and the time commitment of the team members. There was some evidence to suggest that the department head was responsive to requests of the team especially when it was time to provide a timely approval of funds to purchase the new plasma sterilizer.

#### B. SUCCESSFUL IMPLEMENTATION OF BENCHMARKING

How successful has the Materials Management Department at Naval Medical Center San Diego been in implementing

# benchmarking as required by strategic objective 2.5.43 of the 1994 draft of the Navy Medical Logistics Strategic Plan?

As the analysis above shows, the Materials Management Department study shows few of the elements of a benchmarking study. The researcher believes that a lack of understanding of benchmarking may have lead to a miscategorization of the study. Study participants were told to conduct a benchmarking study, but the research indicates the study performed was a cost benefit analysis of two types of sterilization equipment.

A cost benefit analysis is defined as,

A systematic set of procedures by which a firm or government can assess whether to undertake a project or program and, when there is a choice among mutually exclusive projects or programs, which one to undertake. (Stiglitz, 1986, pp. 277-278)

There are two basic approaches to cost benefit analysis: private and social. A private cost-benefit analysis is performed in profit making organizations while a social cost-benefit analysis is germane to not-for-profit organizations. The social cost-benefit analysis was the appropriate method to assess the study performed by the team in the Sterile Processing Division. Social cost-benefit analysis involves,

Determining the broader consequences (inputs and outputs) associated with a project, and the prices at which inputs and outputs are evaluated may not be market prices, either because the inputs and outputs are not marketed (so market prices do not exist) or because market prices do not accurately reflect marginal costs and benefits due to a market failure. (Stiglitz, 1986, p.278) In other words, a social cost benefit analysis is a decision making tool that estimates the cost to society as a result of a prospective policy change. It also tries to assign a monetary value to the benefits an individual may or may not receive due to a prospective policy change. This results in an individual's cost being derived by placing a value on a person's leisure time and life. (Henderson, 1993; Stiglitz, 1986)

Although the Materials Management Department study was successful, the analysis shows that it technically was not a benchmarking study. Therefore, it did not meet strategic objective 2.5.43 of the 1994 draft of the Navy Medical Logistics Strategic Plan.

As described in Chapter IV, the Materials Management Department Head wanted to comply with new federal regulations, minimize employee exposure to known carcinogens and eliminate equipment that polluted the air. Specifically, the analysis was driven by a change to federal regulations and by a desire to minimize the individual's cost in terms of exposure to carcinogenic agents. These reasons are consistent with those noted above for a social cost-benefit analysis.

#### C. RECOMMENDATIONS FOR FUTURE RESEARCH

# 1. Feasibility of Benchmarking in Materials Management Departments in Navy Medicine

Benchmarking has served as a useful tool for many organizations. However, given the multitude of issues confronting Navy Medicine (e.g. capitation budgeting and managed care), what priority should be placed on implementing benchmarking in Materials Management Departments? By conducting a cost benefit analysis, leaders in the medical logistics community can determine some long-

term costs and benefits of the implementation of benchmarking in Material Management Departments.

# 2. Benchmarking Comparison Study

The researcher used a single case study strategy to assess one department's attempt to benchmark. A shortcoming of the approach in this case was that the application of the results were narrowed. Therefore, the researcher advocates the use of a multiple case study strategy, if resources permit. Such a strategy permits a broader application of the study results.

# 3. Assess the Results of the Integration of Benchmarking into a Strategic Plan

When an organization decides to benchmark, they are seeking a competitive edge, world-class status, and to become a leader in customer satisfaction. Integrating benchmarking into the organization's strategic plan is an important step towards attaining those successes as suggested in the literature. Assessing how best practice organizations integrate benchmarking into a strategic plan could assist Navy Medicine in adopting benchmarking as a strong element of their strategy.

#### D. SUMMARY

This chapter has answered the research questions and analyzed the study against the framework of the critical success factors for a benchmarking study. Answers to the research questions were gained by comparing the case study with the definition of benchmarking, the reasons for benchmarking, the Ten-Step Department of the Navy Benchmarking Model and the critical success factors. Although the study did result in enhanced productivity and improved safety and regulatory compliance, it was not a benchmarking study nor was it successful at carrying out

strategic objective 2.5.43 of the Naval Medical Logistics Strategic Plan.

The analysis of the study against the framework of the critical success factors for a benchmarking study resulted in the researcher concluding that the team was not a benchmarking group. The team consisted of people tasked to find the best sterilizer that would meet the needs of NMC San Diego. Benchmarking is a defined process and rarely do managers initiate benchmarking studies in the manner in which this study was done.

# APPENDIX. INTERVIEW QUESTIONS

Date:	Time:
Division:	
Interviewe	ee:
Position:	
Grade:	Rank:
Gender: N	M or F Title: Military or Civilian
GENERAL QU	JESTIONS
1. What d	does the term benchmarking mean to you?
2. Who we	ere the key players in the department's
benchmark:	ing study? What were their responsibilities?
3. How in	nvolved were other members of the department?
Enlisted p	personnel? Civilians?
4. Who le	ed the study? Why was this person selected?
5. Who el	lse was involved in the study? Was there a cross
representa	ation of people in the study group? Was this team
formally a	appointed to do the benchmarking study?
PLANNING_	THE STUDY
1. Why d:	id the department decide to do a benchmarking
study?	
A. 1	Fear of Base Closure
B. 7	Visionary Leadership
С. (	Competition
D. I	Downsizing
0 tibet	arranizations were studied? How were they

2. What organizations were studied? How were they selected?

3. Was a plan of action established before the study?4. Was an external consultant utilized? Who? what were their responsibilities?

#### CONDUCTING THE STUDY

1. How did you conduct the study?

2. What models did you use to guide you through the benchmarking study?

3. How long did it take to perform the study? Should the study have been longer or shorter?

4. What reference materials were used to support you in your study?

5. Who did you consult when you encountered a problem? **LEADERSHIP SUPPORT** 

1. What kind of support did command leadership provide for the benchmarking study?

2. Were there any champions who were not involved in the study? What did they do to assist?

3. Is benchmarking integrated into the command's strategic planning process?

#### OTHER ORGANIZATIONAL EXPERTISE

 Has anyone in the command ever conducted a benchmarking study before?

 Did you seek the assistance of these people before embarking on your benchmarking study? Why or Why not?
Have you seen successful benchmarking studies completed in other departments at this command? What do you think made them successful? How about unsuccessful attempts?
What do you think caused them to be unsuccessful?

#### TQL OFFICE

 Is benchmarking discussed in current Total Quality
Leadership training courses? (Obtain training records and review training materials)

2. What support did you receive from the TQL Office?

#### EDUCATION AND TRAINING

What training did the team receive? Was it on-site?
Was it off-site.

#### LESSONS LEARNED

 Based on your experience with this study, what factors must exist before a benchmarking study is considered a success?

2. What advice would you provide to a department interested in doing a benchmarking study?

3. What were the benefits of conducting the benchmarking study?

A. Would you follow the same procedures as before? Why or why not?

B. What were the strengths of the study?

C. What were the weaknesses or shortcomings of the study?

#### FOLLOW-UP

 Were all of the results of the benchmarking study adopted by management and top leadership? Why or why not?
Would you perform another study? Why or why not?
How long did it take to implement the results of the

benchmarking study?

4. Is the department planning to do future benchmarking studies?

#### REWARDS/INCENTIVES

1. What benefits did the employees gain from the benchmarking study?

2. What awards or recognition were received by the benchmarking team?

3. How does this compare with other awards given at this command?

# PUBLIC RELATIONS

1. Are the success stories publicized?

2. Are benchmarking success stories discussed in the department?
## LIST OF REFERENCES

American Productivity & Quality Center. <u>Applying</u> <u>Benchmarking: A Learning System on CD-ROM</u>. Houston: American Productivity & Quality Center, 1996. .CD-ROM.

American Productivity & Quality Center, Education & Training Division. 1994. <u>Basics of Benchmarking, Organizing &</u> <u>Managing Benchmarking, and Tools & Techniques for Effective</u> <u>Benchmarking Studies Course</u>. Houston: American Productivity & Quality Center.

Beasley, Gary and Cook, Joseph. "The What, the Why, and the How of Benchmarking." <u>Agency Sales Magazine</u> vol. 25, no. 6. June 1995: 52-56.

Blumberg, Donald F. "Strategic Benchmarking of Service and Logistic Support Operations." <u>Journal of Business Logistics</u> vol.15, no. 2. 1994: 89-119.

Bowditch, James L. and Anthony F. Buono. <u>A Primer on</u> <u>Organizational Behavior</u>, 2nd ed. New York: John Wiley & Sons.

Buyson, Ramon. Assistant Department Head of Materials Management, Naval Medical Center San Diego, Personal interview on Benchmarking. San Diego, California, August 26, 1996.

Catabay, Ron. Department Head of Materials Management, Naval Medical Center San Diego. Personal interview on Benchmarking. San Diego, California, August 27, 1996.

Camp, Robert C. <u>Benchmarking: The Search for Industry Best</u> <u>Practices that Lead to Superior Performance</u>. Wisconsin: ASQC Quality Press, 1989.

Cheevers, Jack. "Deep-Sixing CFCs as Ban Looms, Companies and Consumers Feel Pinch." Los Angeles Times, 8 November 1994: Section D.

Conway, Rose. Assistant Department Head of Material Operations, Naval Medical Center San Diego. Telephone interview on Benchmarking. San Diego, California, October 21, 1996.

59

DeToro, Irving. "The 10 Pitfalls of Benchmarking." <u>Quality</u> <u>Progress</u> vol. 28, no. 1. January 1995: 61-63.

Ettorre, Barbara. "Benchmarking: The Next Generation." <u>Management Review</u> vol. 82, no. 6. June 1993: 10-16.

Godfrey, A. Blanton. "10 Quality Trends." <u>Executive</u> <u>Excellences</u> vol. 12, no.7. July 1995: 10-11.

Goff, Vicki. Program Director of the Office of Continuous Improvement, Naval Medical Center San Diego. Personal interview on Benchmarking. San Diego, California, August 28, 1996.

Grayson, C. Jackson, Jr. "Back to the Basics of Benchmarking." <u>Quality</u> vol. 33, no. 5. May 1994: 20-23.

Greengard, Samuel. "Discover Best Practices Through Benchmarking." <u>Personnel Journal</u> vol. 74, no. 11. November 1995: 62-65.

Groom, Dave. Prior Assistant Department Head of Materials Management, Naval Postgraduate School. Personal interview on Benchmarking. Monterey, California, November, 1996.

Haberberger, Cindy. Supervisor of Sterile Processing Division, Naval Medical Center San Diego. Personal interview on Benchmarking. San Diego, California, August 27, 1996.

Henderson, David R. Ed. The Fortune Encyclopedia of Economics. New York: Warner Books, Inc., 1993.

Higginson, Thomas and Robert P. Waxler. "Communication, Commitment and Corporate Culture: The Foundation for TQM and Reengineering." <u>Quality Management</u>. November/December 1994.

Hubbard, Gerald M. "Keys to Creating Performance Measures." Facilities Design vol. 11, no. 5, May 1992: 66-68.

Leibfried, Kathleen H. J. And C. J. McNair. <u>Benchmarking –</u> <u>A Tool for Continuous Improvement</u>. New York. Harper Collins Publishers, Inc., 1992. Leffler, David. Department Head of Sterile Processing Division, Naval Medical Center San Diego. Personal interview on Benchmarking. San Diego, California, August 26, 1996.

Livingstone, Susan. Medical Supply Technician of Sterile Processing Division, Naval Medical Center San Diego, Personal interview on Benchmarking. San Diego, California, August 27, 1996.

"New Techniques needed as CFCs are taken off Market." OR Manager, vol.10, No. 8, August, 1994.

Petreycik, Richard M. "Why Benchmark?" <u>U.S. Distribution</u> Journal (September 1993).

Petrick, Joseph, Scherer, Robert et. al. "Benchmarking and Improving Core Competencies." Journal for Quality vol.17, no. 4. July-August 1994: 82-85.

Sheridan, John H. "Where Benchmarkers Go Wrong." <u>Industry</u> <u>Week</u> vol. 242, no. 6. March 15, 1993: 28-34.

Spendolini, Michael J. <u>The Benchmarking Book</u>. New York: American Management Association, 1992.

Stiglitz, Joseph E. <u>Economics of the Public Sector</u>, 2nd ed. New York: W. W. Norton & Company, 1986.

The Power of Benchmarking: Using Comparative Analysis to Achieve Operational Excellence. Chicago: Foundation of the American College of Healthcare Executives, 1996.

Tomas, Sam. "Benchmarking: A Technique for Improvement." <u>Hospital Materiel Management Quarterly</u> vol. 14, no. 4. May 1993: 78-82.

Tutcher, Gail. "How Successful Companies Improve Through Internal Benchmarking." <u>Managing Service Quality</u> vol. 4, no. 2. 1994: 44-46.

Vasilash, Gary S. "Benchmarking: What It Isn't and Other Observations." <u>Production</u> vol. 106, no. 6. June 1994: 52-53. Venetucci, Robert. "Benchmarking: A Reality Check for Strategy and Performance Objectives." <u>Production</u> vol. 33, no. 4. Fourth Quarter 1992: 32-36.

Watson, Gregory H. <u>Strategic Benchmarking: How To Rate Your</u> <u>Company's Performance Against The World's Best</u>: New York: John Wiley and Sons, Inc., 1993.

<u>Webster's Third New International Dictionary of the English</u> <u>Language Unabridged</u>. Philippines: G. & C. Merriam Co., 1976.

Xerox Corporation. <u>A World of Quality: The Timeless</u> <u>Passport</u>. Wisconsin: ASQC Quality Press, 1993.

Zairi, Mohamed and Huttonn, Rob. "Benchmarking: A Process-Driven Tool for Quality Improvement." <u>TOM Magazine</u> vol. 7, no. 3. 1995: 35-40. Albert, Kenneth J. <u>How to Be Your Own Management</u> <u>Consultant</u>. New York: McGraw-Hill Book Company, 1978.

Boerstler, Heidi, et al. "Implementation of Total Quality Management: Conventional Wisdom versus Reality." <u>The</u> <u>Journal of the Foundation of the American College of</u> <u>Healthcare Executives</u> 41(Summer 1996): 143-159.

Brennan, Niall, ed. <u>Lessons Taught by Baldridge Winners</u>. New York: The Conference Board, 1994. Report Number 1061-94CH.

Brown, Mark G. <u>Baldridge Award Winning Quality</u>, 3rd ed. Quality Resources. Wisconsin: ASQC Quality Press, 1993.

Caro, James L. "Planning and Implementing Total Quality Management (TQM) in a Naval Service Organization: A Case Study of Fleet Numerical Oceanography Center." Thesis. Naval Postgraduate School, Monterey, 1990.

Carvajal, Roberto P. "TQL Implementation Process in the U.S. Navy: A Case Study Onboard the USS Carl Vinson, and its Applicability to Chilean Navy Warships." Thesis. Naval Postgraduate School, Monterey, 1996.

Cohen, Gregory K. "An Assessment of the Implementation of Total Quality Leadership at Marine Corps Logistics Base, Albany, Georgia." Thesis. Naval Postgraduate School, Monterey, 1992.

Evans, Marsha J. Superintendent. Student Guest Lecturer, <u>Seven Rings of Leadership</u>. Monterey, California, June 4, 1996.

Fuller, Robin S. "Modified Benchmarking Study of Program Management Within a Matrix Structure." Thesis. Naval Postgraduate School, Monterey, 1996.

George, Stephen. <u>The Baldridge Quality System: The Do-It-</u> <u>Yourself Way to Transform Your Business</u>. New York: John Wiley & Sons, Inc., 1992.

63

Globerson, Arye, Shlomo Globerson and Judith Frampton. You Can't Manage What You Don't Measure. England: Gower Publishing Company, Limited, 1991.

Henderson, Robert L. "An Analysis of the State of Total Quality in Academia." Thesis. Naval Postgraduate School, Monterey, 1991.

Omachonu, Vincent K. <u>Total Quality Productivity Management</u> <u>in Health Care Organizations</u>. Georgia: Industrial Engineering and Management Press, 1991.

Peters, Thomas, and Robert Waterman. <u>In Search of</u> <u>Excellence: Lessons from America's Best-Run Companies</u>. New York: Harper & Row, Publishers, Inc., 1982.

Russell, J. P. <u>Quality Management Benchmark Assessment</u>, 2d ed. Quality Resources. Wisconsin: ASQC Quality Press, 1995.

Saccomano, Ann. "ISO 9000 Quality Certification Becoming Price of Admission to International Arena." <u>Traffic World</u> April 3, 1995: pp. 50-51.

Scherkenbach, William W. <u>Deming's Road To Continual</u> <u>Improvement</u>. Tennessee: SPC Press, Inc., 1991.

U. S. Department of the Navy. Chief of Naval Education and Training, Code 00Q. <u>The Facilitator's Handbook: A Guide for</u> <u>Effective Presentations</u>. Pensacola, FL, 1992.

Seber, Jim. Management Consulting Firm, Seber Logistics Consulting, Inc.. Telephone interview on Benchmarking. North Brunswick, N. J., June 1996.

Strunk, Jr., William and E. B. White. <u>The Elements of</u> <u>Style</u>, 3rd ed. Massachusetts: Allyn & Bacon, 1979.

Turabian, Kate L. <u>A Manual for Writers of Term Papers,</u> <u>Theses, and Dissertations</u>, 6th ed. Chicago: The University of Chicago Press, 1996.

U. S. Department of the Navy. Navy Personnel Research and Development Center. <u>Fundamentals of Total Quality</u> <u>Leadership</u>. San Diego, CA, 1992. U. S. Office of Personnel Management. Federal Quality Institute. <u>Federal Total Quality Management Handbook</u>. Washington, DC,1992.

.

## INITIAL DISTRIBUTION LIST

1.	Defense Technical Information Center 8725 John J. Kingman Rd., STE 0944 Ft. Belvoir, VA 22060-6218	No. 	Copies 2
2.	Dudley Knox Library	• •	. 2
3.	Total Quality Leadership Office		. 1
4.	Prof. Linda E. Wargo (Code SM/Wg)	• •	. 2
5.	Prof. Erik Jansen (Code SM/Ek)	• •	. 1
6.	LT Pia S. Boston		. 2

. •

DUDLEY KNOX LIBRARY NAVAL POSTGRADUATE SCHOOL MONTEREY CA 93943-5101

